Author's response to reviews

Title: Varicella: epidemiological aspects and vaccination coverage in Veneto Region.

Authors:

Vincenzo Baldo (vincenzo.baldo@unipd.it)
Tatjana Baldovin (tatjana.baldovin@unipd.it)
Francesca Russo (francesca.russo@regione.veneto.it)
Marta C Busana (marta.busana@unipd.it)
Cinzia Piovesan (Cinzia.Piovesan@regione.veneto.it)
Greta Bordignon (greta_bordignon@yahoo.it)
Aurore Giliberti (auroregiliberti@gmail.com)
Renzo Trivello (renzo.trivello@unipd.it)

Version: 2 Date: 8 July 2009

Author's response to reviews:

MS: 3008541722602388
Varicella: epidemiological aspects and vaccination coverage in Veneto Region.
Vincenzo Baldo, Tatjana Baldovin, Francesca Russo, Marta C Busana, Cinzia Piovesan, Greta Bordignon, Aurore Giliberti and Renzo Trivello

Point by point
Referee 1
-The reference #5 has been inserted
-The sections of the manuscript has been appropriately modified as suggested
-Figure 1 and 3 has been deleted and described in the text

Referee 2
-Affiliation #2 has been appropriately translated
-Abstract section has been corrected as suggested
-An appropriate reference has been inserted (Morb Mortal Wkly Rep. 2007,56:1-40) to support the sentence.
-Acronyms has been reported in extenso when cited for the first time (i.e. MMRV and MMR)
-Estimate the number of solicited children has been corrected and better specify. This adjustment has been reported in the method section.
-Punctuation marks has been corrected
-Years were referred to birth cohort. The text has been corrected.
- Sentence has been changed as suggested (section Results, third paragraph, pag. 8)
- In the first paragraph of pag. 10 “12 months” has been corrected with “14 months”.
- All References has been corrected as suggest
- Figure 1 has been deleted (referee #1) and the Figure 2 has been corrected (monthly was indicated in the title of figure)
- Table 2 was corrected with the insert per estenso the type of antigens contain in the vaccine formulation
Discretionary revision
- The “single antigen” has been changed in “monovalent varicella” vaccine as suggested
- Figure 3 and 4 were deleted how suggest and the text was corrected
- Now the discussion section appears more concise.